Author’s response to reviews

Title: Endotracheal intubation with Airtraq versus Storz videolaryngoscope in children younger than two years - a randomized pilot-study

Authors:

Martin K Soerensen (martin@kryspin.dk)
Rolf Holm-Knudsen (rolfhk@gmail.com)

Version: 4 Date: 28 March 2012

Author’s response to reviews: see over
27th of March 2012

Dear Dr Hayley Henderson

Thank you for accepting the manuscript: 2350810556584594.

Title: "Endotracheal intubation with Airtraq® versus Storz® videolaryngoscope in children younger than two years - a randomized pilot-study".

We have made the final changes as you requested.

On behalf of the authors

Yours sincerely,

Martin Kryspin Sørensen, MD